Author's response to reviews

Title: Recurrent Spontaneous Hip Dislocation in Neurofibromatosis Type I: a case report

Authors:

John G Galbraith (johng442@hotmail.com)
Joseph S Butler (josephsbutler@hotmail.com)
James A Harty (jaharty@hotmail.com)

Version: 4 Date: 9 November 2010

Author's response to reviews:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript are now written exactly as they are in the submission system, both in style and order.

Abstract: Abstract in the manuscript matches exactly the Abstract in the submission system

Minor revisions

Email addresses: authors' initials to indicate affiliation to email addresses

Corresponding author: one corresponding author must be indicated on the title page

Structure: Competing interests section now precedes the Consent section

Figures: all mention of figures is consistent all denoted figure